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27 March 2011
MEMORANDUM FOR CENWP-OP, ATTN: Operations Biologist
SUBJECT: Bonneville Lock and Dam, Fishway and Fish Activities for Week 12 of 2011, which covers the period from 20 through 26 March 2011.
1 OPERATION SUMMARY

1.1 Daily average river flows ranged from 204.7 to 246.7 kcfs.  Daily average project tailwater ranged from 18.0’ to 20.0’ msl.  Daily average powerhouse forebay elevation ranged from 72.8’ to 74.2’ msl.  Daily average river temperature ranged from 41 to 43ºF.  Secchi disk measurements ranged from 4.0 to 4.5’.
1.2 Spill Bays 1 and 18, which provide attraction flow for Cascades Island and B-branch, respectively, are open 6” during daylight hours.  Daily average spill was 1.4 kcfs during the week.

1.3 Unit Operation.  Powerhouse Two is the priority powerhouse.
1.3.1 Unit 11 has remained out of service since 12 June 2008 due to rotor damage.
1.4 Second Powerhouse fish units provide attraction flow for the Washington Shore (WS) fish ladder.  The units may be placed on standby, as necessary, during nighttime hours to float debris.
      1.4.1
    F2 was out of service from 0040 until 0712 on 22 March to work on breaker. 

1.5 Gatewell drawdowns are measured on Wednesdays.
2 ACTIVITIES

2.1 Auxiliary Water System Closures.  The following AWS valves were closed this reporting week:
2.1.1 FV 3-9 was closed for cleaning from 0706 until 0730 on 24 March. 
2.2 STS/VBS inspections.  No STS/VBS inspections occurred this reporting week.
Table 1.  STS and VBS inspections.  
	Unit
	Date
	Run Hours
	Comments

	11
	NA
	OOS
	U11 remains out of service.

	12-18
	NA
	NA
	Inspection scheduled for week of 04 April


2.3 Dewatering and Fish Salvage.  The following fish salvage activity occurred this week:

2.3.1 On 20 March, Project Fisheries salvaged 2 yearling chinook from the upstream sill of the Navigation lock.  They were released in good condition at the Hamilton Island boat launch.

3 RESEARCH  
3.1 Battelle - survival study.  Acoustic tag training began the week of 07 March.  Fish for training currently held in raceways at the SMF.  Hydrophone installation continues.
3.2 CRITFC - Sea lion hazing.  Hazing from boats continues.
3.3 FFU - Sea lion.  Observations continue.

3.4 NOAA - lamprey. DIDSON mount installed on the NDE LPS on 15 February.  
3.5 ODFW/WDFW - Sea lion trapping. Sea lion traps remain adjacent to the B2CC and downstream of the old Navlock.  Trapping has not begun for the season.
3.6 PSMFC.  Juvenile fish condition and index monitoring continues.  Fallbacks recorded crossing the separator bars:  13 steelhead and one unknown salmonid.  Please see http://www.fpc.org/adultsalmon/bonkeltcounts.htm for updated kelt counts.
3.7 WDFW - Fish Counting.  Video counting continues.   Visual counting scheduled to begin 01 April.
3.8 USDA - Sea lion hazing. Hazing from the deck continues.
4 FISHWAYS 
4.1 Project biologists inspected 20 through 26 March.
Fish Passage Plan Observations.

4.2 Table 2.  FPP Out of Criteria items.
	DATE
	LOCATion
	FPP itEMS
	CAUSE
	RESPONSE
	TR#

	BRADFORD ISLAND (BI) 

	3/20
	B-branch entrance
	Differential  > 2.0’
	FV 4-3, 4-4 in manual
	Operations adjusted.
	

	3/22
	PH1CC north/south entrance
	Differential < 1.0’
	FV 1-1 in manual
	Operations adjusted.
	

	CASCADES ISLAND (CI) – nothing to report

	WASHINGTON SHORE (WS) – nothing to report


4.2.1 PH1 PCC diffuser FG2-19 is partially open.  Repairs scheduled for 2011/2012 in water work period.  

4.2.2 The PH1 PCC mechanical velocity meter has been repaired and a new mechanical velocity meter has been fabricated for PH2. Both are scheduled for installation.
4.2.3 The SA-24 board that controls the Powerhouse 2 adult ladder components is in need of replacement. Work has begun on replacing the PLC to allow for fully automated operation of the PH2 adult fishway.
4.3 Fishway Temperatures.  Temperatures are taken 01 March through 30 November.

Table 3.  Bonneville Fishway Temperatures

	Date
	WA. Shore average
	BI

average

	3/20
	40.7
	41.2

	3/21
	40.8
	41.3

	3/22
	41.2
	41.6

	3/23
	41.4
	41.8

	3/24
	41.4
	41.8

	3/25
	41.4
	41.9

	3/26
	41.7
	42.2


4.4 Zebra Mussel Monitoring.  No signs of colonization were observed this reporting week.  
4.5 Adult Fishways. 
4.5.1 Calibration was checked on 26 March.
4.5.2 10 of 12 floating orifice gates (FOGs) are installed with sea lion barriers in place.  Two gates remain out for maintenance.

4.5.3 SLEDs remain installed at all fishway entrances.
4.6 Juvenile Fishways.
4.6.1 DSM2 remains in service.
4.6.2 The PH1 ice and trash sluiceway (ITS) remains in service.  Manual chaingates 1A and 1B remain open. Automated chaingates 3B, 6C, and 10B are open and functioning in auto.
4.6.3 The hydrocannon remains in service.
4.6.4 The B2CC remains in service.
4.7 Lamprey Fishways.
4.7.1 The Cascades Is. entrance weir remains installed.  Cascade Is. LPS remains out of service.
4.7.2 The NDE ramp remains in the lowered position and is not currently operating.
4.7.3 Bradford Island and WA Shore LPS’s remain out of service.
5 CONSTRUCTION
5.1 PH1 Gantry 4.  New crane in June, 2011.  
5.2 Spillway Gate Repair Pit. Clean-up of hazardous material and shoring of the walls is complete.  Load test scheduled to occur spring 2011.
5.3 B2CC Hoist. Construction scheduled to begin fall of 2011.
6 HAZMAT, SPILLS AND CLEANUP. Nothing to report.
7 TELETYPES AND OTHER NOTEWORTHY ITEMS
7.1 A teletype sent on 22 December outlines new tailwater operations for chum incubation.  Effective immediately, BON is requested to maintain a tailwater of 12.2’ or greater for all hours.

7.2 A teletype sent on 22 December notifies BON that slow roll start-up for units that have been idle for 12 hours or more is required from 01 December through 30 April (see FPP section BON 5.7).Note: Though specified in the FPP and the teletype as only occurring when units have been idle for 12 or more hours, Bonneville has been starting all idle units with slow roll protocol.  
7.3 A teletype sent on 15 March authorizes BON to open the corner collector (B2CC). If TDG at the Warrendale site exceeds 110% after 24 hours of operation, close Spillbays 1 and 18.  If the 110% TDG is still exceeded after 24 hours operating without spillbays, then close the B2CC and operate PH1 as the priority powerhouse.

7.4 A teletype sent on 22 March outlines a short term closure of the B2CC from 0700 to 1200 on 23 March to repair a damaged trolley pipe at the B2CC entrance.
7.5 A teletype sent on 24 March outlines a short term closure of the B2CC from 1200 until 1700 on 29 March to remove a damaged trolley pipe at the B2CC entrance.

7.6 A teletype sent on 24 March authorizes BON to close the PH1 ITS from 0700 until 1700 each day on 28 and 29 March.  This is operation is necessary to support relocation of three trolley pipes in the PH1 forebay.
GLEN A. SMITH, P.E.
Operations Project Manager
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